Author's response to reviews

Title: Where do students in the health professions want to work?

Authors:

Deborah Schofield (deborah.schofield@ncahs.health.nsw.gov.au)
Susan Fletcher (deborah.schofield@ncahs.health.nsw.gov.au)
Jeffery Fuller (jeffrey.fuller@ncahs.health.nsw.gov.au)
Hudson Birden (hudson.birden@ncahs.health.nsw.gov.au)
Sue Page (sue.page@ncahs.health.nsw.gov.au)

Version: 4 Date: 29 June 2009

Author's response to reviews: see over
29 June 2009

Dr Mario Dal Poz
Editor-in-Chief
Human Resources for Health
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK

Dear Dr Dal Poz

Re: Revision of “Where do students from different disciplines want to work?”

We have made the following revisions to the paper “Where do students from different disciplines want to work?”:

- It has been noted in the Authors Contribution section that all authors have read and approved the final manuscript;
- Reference details have been updated to include all the required information;
- The order of the references in the document text has been revised; and
- The results section has been renamed to ‘Results and Discussion’.

Best regards

Deborah Schofield